Author’s response to reviews

Title: Folate, vitamin B12, ferritin and haemoglobin levels among women of childbearing age from a rural district in South India

Authors:

Samiksha Singh (samiksha.singh@iiphh.org; drsamikshasingh@gmail.com)
Jaga Jeevan Babu Geddam (geddambabuj@yahoo.com)
G. Bhanuprakash Reddy (geereddy@yahoo.com)
Dinesh Raj Pallepogula (p.dineshraj@gmail.com)
Hira Ballabh Pant (hira@iiphh.org)
Sutapa Bandyopadhyay Neogi (sutapa.bneogi@iiphd.org)
Neena John (neena.sushil@iiphh.org)
Sunanda Reddy Kolli (write2sunanda@gmail.com)
Pat Doyle (pat.doyle@lshtm.ac.uk)
Sanjay Kinra (sanjay.kinra@lshtm.ac.uk)
Andy Ness (andy.ness@bristol.ac.uk)
Gudlavalleti Venkata Satyanarayana Murthy (murthy.gvs@iiphh.org)

Version: 3 Date: 16 Jun 2017

Author’s response to reviews:

Dear Editor,

Thank you for accepting the manuscript with some editorial changes. We have accepted all your suggestions and edited appropriately. The changes are highlighted in yellow for your reference.

Regarding availability of data: The data is not available in public domain however it could be requested. We have added appropriate statement in the manuscript.
Regarding authorship: We by error omitted some description in authorship contribution. We have now clarified their roles and involvement more elaborately in the Author contribution section. We wish to retain all the authors as all qualify the authorship criteria of your journal.

Author's contribution

Conception of the study: GVSM, JJBG, GBR, SBN, SRK, PD, SK, SS and AN. Data analysis and interpretation: SS, DRP and HBP. Drafting the manuscript: SS, NJ, DRP, HBP, JJBG, and GBR. Critical revision of the Manuscript: GVSM, PD, AN, SK, SKR, and SBN. Writing of the final manuscript: SS, JJBG, GBR, GVSM and DRP. Final Approval of the version to be published: GVSM, AN, SK, PD, SRK, SBN, NJ, HBP, DRP, GBR, JJBG and SS.

Thanks for your editorial support.

Samiksha